Author's response to reviews

Title: Factors associated with post-traumatic stress disorder and depression amongst internally displaced persons in northern Uganda

Authors:

Bayard Roberts (bayard.roberts@lshtm.ac.uk)
Kaducu Felix Ocaka (fkaducu@yahoo.co.uk)
John Browne (john.browne@lshtm.ac.uk)
Thomas Oyok (oyokthomas@yahoo.co.uk)
Egbert Sondorp (egbert.sondorp@lshtm.ac.uk)

Version: 7 Date: 19 May 2008

Author's response to reviews:

Dear Sir/Madam,

Further to the email of 15 May 2008, the manuscript has been re-submitted (entitled: Uganda IDP mental health 190508). The following changes were made to the manuscript. These are generally typographical in nature and none are substantive.

Line 2, paragraph 1 of p.4: Adding ‘s’ to ‘layout’.
Line 3, paragraph 1 of p.4: Adding ‘s’ to ‘camp’.
Line 13, paragraph 1 of p.4: Removing ‘of which 227,245 were of eligible age’ as felt not to be necessary.
Line 1, paragraph 3 of p.4: Removing the ‘ ’ from pre visit.
Line 3, paragraph 3 of p.5: Changing ‘6’ to ‘Six’.
Line 3, paragraph 1 of p.7: Changing ‘over 40% to ‘41%’.
Line 2, paragraph 1 of p.9: Changing ‘indicate’ to ‘indicates’.
Line 3, paragraph 1 of p.9: Removing ‘The study provides evidence on the abduction of civilians in northern Uganda’.
Abbreviations: ‘Odds Ratio (OR)’ moved to before ‘Post-traumatic stress disorder (PTSD)’.

The three figures that were reformatted by Biomed Central look fine and no further changes are required.

I consider this re-submitted manuscript ready for publication.

Yours sincerely. Bayard Roberts